Dear editors,

Thank you very much for sending us the reviews and considering our manuscript for publication.

The title has been changed as proposed by the editors.
In the following we respond point by point to the reviewers' comments:

Reviewer David Hardisson:

Methods: Follow-up included MRI or CT of the tumor site and abdominal ultrasound (changes in the abstract and text were made accordingly). Detailed information of the antibodies is now provided in table 1 (which can be deleted by the editors).

Results: The error in tumor size was corrected. The fact, why an intraoperative incision of the tumor should be considered an R1 resection is now explained.

Discussion: The first sentence is now changed as well as the order within the first paragraph. A reference number is now indicated.

Reviewer Giuseppe Cardillo:

The first sentence of the discussion is now changed. The issues of radiotherapy and its possible benefits as well as the role of PET scan are now addressed. A life long follow-up is recommended.

The manuscript has now been corrected by a native speaker.

We hope our changes in the manuscript make it worthy for publication in your journal. Thank you very much for your kind consideration.

Sincerely,

Dr. med. Adrien Daigeler